Notes of the RICH1 Phone meeting 02/02/2007
People Present:
Bristol: Nick(part), Richard

CERN: Fabio, Kenneth(part), Olav(part), Richard P.
IC:        Bill, Dave, Gabriel(part), Neville
Oxford: Matthew
Apologies for absence:
Mike,  Rohan
Discussed Agenda Topics:
1. Gas Enclosure
When the beampipe is aligned, the seal to the exit window can be completed (action Derek/Peter), the exit window bolts tightened and the leak tests can be done (~2 weeks).

2. Quartz Windows
All 4 QW have been coated.  The transmission has been measured by Kenneth to be ~96%. The 4 QW are safely stored awaiting installation later on in the year.
3. Spherical Mirrors
CMA: At last Wednesday’s CMA phone meeting the packaging and handling of the CMA box (mirrors+frame) was discussed. CMA warned us to pay special attention when handling the mirrors because they are fragile i.e. skins can detach from core if some force is exerted. The mirrors fragility is comparable to the demo mirror. SESO will be warned to pay special attention when handling the mirrors. The packaging, transport, cleaning and coating was discussed. CMA is content with the proposed coating plans at SESO.  CMA will send all the necessary documentation (.pdf files) next week.  We will remain in contact with CMA and let them know how things are progressing: D0,R measurements of the 4 mirrors, coatings at SESO etc. The CMA wooden box (mirrors+frame) weighs ~180kg and is expected to arrive in ~1 week.  Once the box is at CERN it will be brought to the clean room in Bat. 156. The mirrors will be measured (D0,R)  (action Fabio/Kenneth) and packaged for transport to SESO (action Fabio/Andre).
C4F10 Testing: On the 22 January 07 the fluorocarbon gas tank was opened and the samples were tested. The exposure time to C4F10 is 7.5 months for Flat Sample A, Mirror Sample 1 and 6.5 months for the demo mirror. The series of tests made (optics, mechanical and metrological) give no indication of any change in the properties of the samples. The samples are back in the tank and will be tested in ~4 months.

Coating at SESO:  The mirrors will be brought to SESO by van (action Fabio) for coating during week 8. SESO has agreed to the planned delivery during week 8. If there is any delay the mirrors can be coated by SESO not before the end of March07. The CERN finance office has stopped the order for coating the mirrors (SESO Step 2) because it is not in accordance with CERN rules. The enquiry should have been done through CERN which would have required much more time. Werner has written a justification.  (After meeting: green light given by CERN finance 5Feb07).
Mirror Installation tooling: Drawings completed. The tooling is being manufactured. Delivery ~ end Feb07
Mechanics: The V-blocks need to epoxied in the carbon fiber feet. It will be done at CERN when the alignment rig also at CERN (action Geoff).
4. Flat Mirrors
Optical alignment setup: Matthew has completed the model design. The manufacturing has started and the rig should be ready by end of February. The drawings are in EDMS. 
.
Installation Boxes: Richard altered the shelf plate to allow more adjustment to accommodate the 100 mm of the extended top plate of the rig. The slight discrepancies can be adjusted with shims. Richard’s updated drawings were given to the CERN “Sous Traitance” service. CERN “Sous Traitance” received three quotes this week.  At the meeting the decision was taken to accept the 15k CHF offer with delivery date mid March07. The order was sent out in the afternoon after the meeting (2/2/07).
5. HPD Assembly
Mechanics: The design of the HPD box is finalized and the material ordered. When material arrives, in house manufacture can begin. All aluminium ladders are done.  The cooling plates are being manufactured and 8 plates have been assembled.

Dressing: HPDs have been mounted on the first RICH1 column.  Low voltages have been set. Next week the HV will be put on and the test patterns tested. 

Shields: The RICH1 Mumetal shields state is (e-mail from Barbara):

205 tubes rolled with Kapton, tubes from 1 to 167 already tested ,1 tube rejected (041).

Rail System:  Last week the model for the frame+hood and rail system have been married together. Small problems were identified and Rohan is making minor modifications. Do not yet have the quotes for manufacture.

Cabling: Cabling is coordinated by Ken and Rolf. The cable trays from the balcony to the lower part of RICH1 have been installed, and arrive at x=0 (edge of concrete) while the cable trays to the top of RICH1 stop at the tunnel and will be extended when the scaffolding is mounted for bringing in the cables. The cables are connected in situ.   
Photon Funnel: This object provides many functions: Interface to HPD Chariot, Frame for QW, Compliance between GE and HPD chariot and MDMS.

6.  MDMS
IC-Syracuse meeting this afternoon 2/2/07. 
7. LAMS

The fibers are laid down from counting room to balcony. The 3 remaining cameras have been bought. The focalizers need to be bought (action Gabriel). Next week Gabriel will meet with Antonis at CERN. 

.
8. Aerogel
No news.
9. Next meeting
Friday 16th February 2007 at 10:00(GMT) 11:00(CET). 
